Author's response to reviews

Title: Asthma in Black African, Black Caribbean and South Asian adolescents in the MRC DASH study: a cross sectional analysis

Authors:

Melissa J Whitrow (melissa.whitrow@adelaide.edu.au)
Seeromanie Harding (seeromanie@sphsu.mrc.ac.uk)

Version: 5 Date: 16 January 2010

Author's response to reviews:

Please find attached the final proof read version of our paper entitled "Asthma in Black African, Black Caribbean and South Asian adolescents in the MRC DASH study: a cross sectional analysis".

Yours sincerely,

M Whitrow
S Harding